Author’s response to reviews

Title: Case Report: A fatal case of disseminated adenovirus infection in a non-transplant adult haematology patient

Authors:

Michael Joffe (Michael.C.Joffe@uhl-tr.nhs.uk)

Simon Wagner (Simon.Wagner@uhl-tr.nhs.uk)

Julian Wei Tze Julian Wei Tze (jwtang49@hotmail.com;jwtang49@gmail.com)

Version: 4 Date: 15 Jan 2018

Author’s response to reviews:

1) Ethics approval and consent to participate: since the patient was treated according to standard care, please write here "Not applicable". Please remove all wording concerning consent to publish, as this is not the correct place to provide this information.

   - this has been amended, as requested.

2) Consent to publish: this refers to the consent obtained from patients to publish their individual data, and not the consent from authors to submit their manuscript. Please write here: "Written consent to publish this case report and any accompanying images was obtained from the patient’s next of kin". Please remove wording concerning authors' consent, which should be moved to the end of the Authors contributions section instead.

   - this has been amended, as requested.

3) Authors’ contributions: at the end of this section, please write "All authors have read and approved the final version of this manuscript"

   - this has been amended, as requested.

4) Competing interests: please rephrase this section so that it reads "JWT is one of the Virology Section Editors for the journal. The other authors declare that they have no competing interests."

   - this has been amended, as requested.

5) Please remove all additional files, as these are no longer needed. If you wish to provide a point-by-point response to these requests, please do so using the appropriate box during submission.
- this has been amended, as requested.